Author’s response to reviews

Title: Chronic Hepatitis Infection Is Associated with Extrahepatic Cancer Development: A Nationwide Population-Based Study in Taiwan

Authors:
Abram Kamiza (abramkamiza@yahoo.co.uk)
Fu-Hsiung Su (williamsufh1@yahoo.com.tw)
Wen-Chang Wang (wangwc@tmu.edu.tw)
Fung-Chang Sung (fcsung1008@yahoo.com)
Shih-Ni Chang (u9665856@cmu.edu.tw)
Chih-Ching Yeh (ccyeh@tmu.edu.tw)

Version: 2 Date: 26 Oct 2016

Author’s response to reviews:

Rudolf Kaaks,
BMC Cancer
October 26, 2016
Re: Paper No: BCAN-D-16-01507
Titled: Chronic Hepatitis Infection Is Associated with Extrahepatic Cancer Development: A Nationwide Population-Based Study in Taiwan

Dear Professor Rudolf Kaaks:

We thank you for your inspiring and helpful comments and suggestions for our manuscript. We have carefully revised the manuscript as suggested and have provided our point-by-point responses to the reviewers’ comments. We thank you for considering this article for publication in BMC Cancer.
Sincerely,
Chih-Ching Yeh, PhD, MSPH
Professor,
School of Public Health,
College of Public Health and Nutrition
Taipei Medical University
250 Wu-Hsing Street,
Taipei City 110, Taiwan
E-mail: ccyeh@tmu.edu.tw
Phone: +886-2-2736-1661 ext. 6534
Fax: +886-2-2738-4831
Responses to Reviewers’ Comments

Paper No: BCAN-D-16-01507
Titled: Chronic Hepatitis Infection Is Associated with Extrahepatic Cancer Development: A Nationwide Population-Based Study in Taiwan

Reviewer reports
Please remove the "Electronic word count" section from the manuscript.
Reply: Thank you for your thoughtful suggestion. In the revised manuscript, we have removed the electronic word count.

Please move the "Abbreviations" section to follow Conclusions and rename to "List of abbreviations".
Reply: Thank you for your thoughtful suggestion. We have renamed the abbreviations to list of abbreviations after the Conclusions section, line 17, and page 16 of the revised manuscript.
Please rename the "Financial support" declaration to "Funding".

Reply: Thank you for your suggestion. We have renamed the financial support to funding in the Declaration section, line 29, and page 17 of the revised manuscript.

When resubmitting, please provide a clean copy of the manuscript with no colored text or other markup.

Reply: Thank you for your thoughtful suggestion. In this revision, we have submitted a clean copy of the manuscript.